Author's response to reviews

Title: Filter presence and tipping paper color influence consumer perceptions of cigarettes

Authors:

   Richard J O'Connor (richard.oconnor@roswellpark.org)
   Maansi Bansal-Travers (maansi.travers@roswellpark.org)
   K. Michael Cummings (cummingk@musc.edu)
   David Hammond (david.hammond@uwaterloo.ca)
   James F Thrasher (THRASHER@mailbox.sc.edu)
   Cindy Tworek (ctworek@hsc.wvu.edu)

Version: 2 Date: 13 January 2015

Author's response to reviews:

Per the editors request, we have made the following changes:
1. Added line numbering and page numbering
2. Reformatted author contribution
3. Added acknowledgements section
4. Adding IRB approval statement